

Response letter

Reviewer 1:

This a well documented and written case report and review of the literature. I suggest that it would be useful to include in the conclusions that this rare complication should be included in the preoperative surgical consent.

Response:

Thank you. A statement was added in the conclusions.

Reviewer 2:

Short work was concise and clear. Please consult a proofreader for the English language Some references are old and should be replaced by newer.

Response:

Thank you. The article has been revised and the language corrected by a proofreader. A literature review is made considering all available cases of hepatic hematoma after ercp. Older references are consistent with this effort and included in Table 1.

Thank you for you precise revision.

Sincerely

Prof. MA Zappa, MD